Author’s response to reviews

Title: Polio Eradication Initiative in Africa: Influence on Other Infectious Disease Surveillance Development

Authors:

- Peter Nsubuga (pnsubuga@cdc.gov)
- Sharon McDonnell (sem0@cdc.gov)
- Bradley A Perkins (bap4@cdc.gov)
- Roland W Sutter (rsutter@cdc.gov)
- Linda Quick (mag2@cdc.gov)
- Mark E White (mhw1@cdc.gov)
- Stephen Cochi (scochi@cdc.gov)
- Mac Otten (ottenm@whoafr.org)

Version: 4 Date: 9 Dec 2002

Formatted and revised according to the e-mail from Dr. Collett on December 2, 2002